RESEARCH ARTICLE

Open Access

Hearing the voices of older adult patients: processes and findings to inform health services research



Sally Fowler Davis^{1*}, Anne Silvester², Deborah Barnett², Lisa Farndon² and Mubarak Ismail³

Plain English summary

Whilst Patient and Public Involvement and Engagement (PPIE) are widely regarded as critical to developing clinical research, there is a perception that older adults may not be able to contribute and there is less emphasis on gaining a wide range of opinions before developing research questions or projects; for example an organisational change. This PPIE initiative used three PPIE processes including existing panels and wider networking to access older adults in the community who had used the hospital services and been discharged. Older adults expressed a range of views about their experience of discharge planning and this provided an important perspective on patients' research priorities associated with their personal independence. Efforts were taken to ensure representative views across a cross section of the population. As a result of this initial PPIE, a permanent, co-ordinated 'elders' panel has been established to ensure a representation of older adult views for research, service development and evaluation. This panel has permanent, fully supported members who provide reflection and feedback on any projects and programmes relating to older people's services in the City.

Abstract

Background Clinical academic research and service improvement is planned using Patient and Public Involvement and Engagement (PPIE) but older PPIE participants are consulted less often due to the perception that they are vulnerable or hard to engage.

Objectives To consult frail older adults about a recently adopted service, discharge to assess (D2A), and to prioritise services improvements and research topics associated with the design and delivery of discharge from hospital. To use successive PPIE processes to enable a permanent PPIE panel to be established.

Participants Following guidance from an established hospital PPI panel 27 older adult participants were recruited. Participants from Black, Asian and Minority Ethnic (BAME) communities, affluent and non-affluent areas and varied social circumstances were included.

Methods Focus groups and individual interviews were conducted in participants own homes or nearby social venues.

(Continued on next page)

¹Sheffield Hallam University and Sheffield Teaching Hospitals NHS Foundation Trust, Montgomery House, 32 Collegiate Crescent, Sheffield S10 2RR LIK

Full list of author information is available at the end of the article



^{*} Correspondence: s.fowler-davis@shu.ac.uk

(Continued from previous page)

Results Priorities for discharge included remaining independent despite often feeling lonely at home; to remain in hospital if needed; and for services to ensure effective communication with families. The main research priority identified was facilitating independence, whilst establishing a permanent PPIE panel involving older adults was viewed favourably.

Conclusions Taking a structured approach to PPIE enabled varied older peoples' voices to express their priorities and concerns into early discharge from hospital, as well as enabling the development of health services research into hospital discharge planning and management. Older people as participants identified research priorities after reflecting on their experiences. Listening and reflection enabled researchers to develop a new "Community PPIE Elders Panel" to create an enduring PPIE infrastructure for frail older housebound people to engage in research design, development and dissemination.

Keywords: Frail older adults, Patient and public involvement and engagement (PPIE), Discharge planning, Patient experience,

Introduction and background

This paper reports on a community-based initiative to enable the voice of frail older people to inform research and service development within the National Health Service (NHS) in England. A frail older adult in this paper refers to a person over 75, with some mobility issues and a wide range of health conditions and care needs. Patient and Public Involvement and Engagement (PPIE) in health and social care can be for the purpose of planning research and to understand the priorities of service users and here, both approaches were used [1]. The goal was to understand patient experience to generate research that improved services and produced better patient outcomes [2]. Involvement and engagement of a wide variety of older people was sought in a large provincial city in England, across a range of community settings. Group and individual meetings were arranged via key contacts and links with representative community leaders to access affluent and non-affluent areas as well as people from Black, Asian Minority Ethnic (BAME) Communities.

Consultation with a growing older adult population is an important agenda for the National Health Service (NHS) in terms of research and for the purpose of service development. A series of PPIE activity enabled successful recruitment in terms of numbers and the diversity of those consulted and a permanent PPIE group to consult on future research design and processes.

The acknowledged benefits of consulting older people include facilitation of health technologies, illumination of areas of practice leading to improvements in care, enhancement of both the quality of research and its potential for helping address more intractable health problems, and improving the chances of more relevant research being conducted [3–8]. Frail older house-bound people can unwittingly be excluded from consultation because, for practical reasons, they are less accessible. A further possible constraint to community participation

is those being most sought and required are often "least likely to be in a position to donate their time and energy" [8]. The literature also suggests that traditional PPIE panels are often conducted via focus groups [5, 9], but the population of frail older people (particularly aged 80 and over) may have multiple health, mobility or vision difficulties and are unlikely to respond well to written questionnaires, telephone consultations or one off group meetings. Consensus on the preferred method of consultation with older people is for individual face-to-face home interviews [10]. It is suggested that using gatekeepers such as local clinicians and/or community leaders to access more commonly excluded groups, such as older adults, can be a helpful means of recruitment and developing trust [5, 11].

Greater involvement of users, carers and service providers in the design, delivery and evaluation of research associated with services for older people is now widely promoted [7] with a corresponding need to enable robust consultation and communication about critical issues in health and care planning. This paper describes how frail older people were engaged in discussions about discharge from hospital, identifying their issues and concerns and leading to a detailed understanding that could be used to focus research and service change. This initiative resulted in an ongoing Community Elders PPIE Panel.

Health services for older adults

The frail older population comprise an increasing proportion of the United Kingdom (UK) population with 17.7% of adults aged over 65, a figure that is set to rise [12]. Within Europe those aged 80 and over will be the fastest growing age group, increasing from 5 to 12% between 2010 and 2060 [13]. Previous research has indicated that those who are frail and older are more at risk from adverse health outcomes, making it important that future research is conducted to improve and identify outcomes that are both health improvements and service improvements [5, 14].

An increasing problem in regard to frail older patients is delayed discharge from hospital, a trend that is rising. This is illustrated by the National Audit Office who calculated a 31% rise in delayed transfers of care between 2013 and 2015, with an estimated £820 million cost to the NHS of older patients in hospital beds who are no longer in need of acute treatment [15]. Longer hospital stays can lead to worse health outcomes, particularly relevant to the older adults, who also have the additional likelihood of increased long term care needs [15]. Acknowledged adverse outcomes from delayed hospital discharge include medical complications, infections, loss of mobility, reduced capability of carrying out daily living tasks, or even death [15]. In addition they can have a negative effect on patients' health after discharge, increasing their chances of readmission to hospital [16]. Frail older patients also express their desire to be discharged in a timely manner, although with the caveat that they are physically well enough and have sufficient support at home to be able to cope [17].

Health service research and the changing models of care promoted by NHS England (NHS England 2015) have produced recommendations that discharge planning should begin as soon as possible after older adult patients are admitted into an acute hospital bed. Furthermore, a "discharge to assess (D2A)" model should be implemented, where older patients are assessed for care and equipment needs at home, rather than waiting in hospital for rehabilitation and social care assessment. Thus hospitalised older patients receive less therapy in hospital [18] and return home to receive community care [19]including nursing, therapy and home support. It is suggested this is the default pathway, with alternative pathways for those who cannot return home immediately [20].

Funding for patient and public involvement and engagement

In July 2014 a Patient and Public Involvement and engagement (PPIE) project was funded with a small grant awarded by the Yorkshire and Humberside Research Design Service, National Institute for Health Research (NIHR). This award enabled access to an established PPIE group to advise on accessing a range of older patients' voices regarding D2A, alongside supporting the preparation of a clinical fellowship application (see NIHR Clinical academic career fellowships) [21, 22]. This project was the first initiative to seek patient and public views into discharge management in the city wide D2A service and incorporate these into future improvements.

Ethics and governance

As a PPIE initiative, this project was deemed to be exempt from ethics by the clinical research office in the sponsoring NHS Trust. The work was registered as a service review and carried out by clinical academics working within the Trust, seeking the views on elder research and furthering the development of a sustainable PPIE infrastructure. The funders who are Regional Design Service of the NIHR in Yorkshire and Humber (RDS Y&H) recognised PPIE as service review activity.

Methods

Design

An established NHS PPIE group, consisted of patient representatives from the local area and a co-ordinator post was hosted in a research team. The PPIE members were experienced in advising researchers about the methodology and burden of research activities, so they were approached. The PPIE panel were advised of the primary problem; how to access frail older participants who had experience of ward services and were willing to reflect on their experiences of discharge from hospital. The PPIE panel recommended that due to the limited funding of the £500 NIHR grant, the researchers should approach existing groups, social clubs and networks that attracted regular participation from frail older adults. Based on this advice, existing and known groups across a range of demographics were targeted to seek diverse members of the older adult population, including those whose opinions are rarely sought. The inclusion of different groups from BAME backgrounds and from deprived communities and those who may be less sociable or more isolated became part of the formal plan. The goal was to identify eight frail older participants who by virtue of a recent admission or their family carer role could provide a range of opinion into medical and care needs on discharge.. Beyond this discharge project, the initial PPIE activity was to be extended to involve older adults as stakeholders within the city's research community, helping to direct and prioritise research more widely.

Recruitment

Representatives of community groups and specific contacts were asked to identify any older members who may be interested in taking part in a PPIE focus group or interview in order to talk about their experiences of hospital discharge, especially D2A, and priorities around discharge. Using gatekeepers such as local clinicians or community leaders to access more commonly excluded groups was seen as a helpful means of recruitment and developing trust [5, 18]. Interested parties were provided with an information sheet about the project, which included the topic questions which were to be used for the discussions (Appendix). Volunteers who agreed to take part were given an appointment to talk individually or be part of a focus group in a place of their choice. Most preferred to be visited at a community venue but not a

hospital site with just two preferring to be in their own home. Individuals and groups were invited to participate with support for travel, refreshments and any untoward costs associated with room hire. This enabled small organisations to facilitate by inviting members and potential participants and offering hospitality on behalf of the PPIE initiative.

Participants and attention to diversity

Contacts and groups approached included a retirement village, a BAME community group, a lunch club in a non-affluent area and an over 50s club in an affluent area. Nine older female participants were recruited from the retirement village and agreed to take part in a focus group in one of their community rooms. A researcher colleague with a prior relationship with the Pakistani Muslim Centre (PMC) approached the current manager who helped to identify thirteen men of Pakistani, Yemeni and Somali origin willing to take part. Assistance was provided by PMC leaders to enable consideration to be given to language and methods of asking for informed consent. The group took place within the PMC to ensure familiarity and engender trust. A chair of an over 50's club in an affluent area was approached and facilitated a discussion within a local church hall. Although more than 30 people were present, only three women actively participated. Finally two members from an established lunch club in a non-affluent area volunteered to home-based interviews.

Participant focus groups and interviews

To ensure common areas of questioning and to explore specific issues a list of topic questions guided the researchers who facilitated the groups or individual discussions [14] (Appendix). It enabled participants to share their experience and reflection on hospital care and need for supported discharge, and focus possible future research questions derived from service user experience. Two people facilitated each group or individual meeting to enable effective note taking. The notes were shared at the end of each meeting and participants were then provided with contact information, and further invited to be involved in more formal PPIE panels if they wished.

Collating findings and presenting stakeholder involvement

Following the meetings all notes were collated and discussed between researchers to determine the main findings about discharge and to draw conclusions from the suggestions and experiences, particularly in relation to the priorities that suggested further research focus. Further reading and analysis was undertaken to fully understand and to appreciate any differences and similarities between the views of individuals and groups, in addition to formulating priority issues as identified by the

participants. The NIHR PPI report was formulated and presented to the NIHR in April 2015 further NIHR proposals have been submitted; based on the understanding of frail older adult priorities reported in this study.

Findings

In total 27 individual responses were collected. The experiences of frail older NHS patients on admission and discharge are summarised as a series of topics to inform the continuing service improvements and to ensure that research planning was prioritised around patient experience. They are as follows:

Priority 1: Reducing hospital length of stay

All participants appreciated the need for hospital stays to be as short as possible and if they were well enough they would prefer to be at home, even when their circumstances were not ideal. They agreed that continuing assessment at home was preferable to longer stays. Staying in hospital was also considered a problem because of the costly travelling and parking if patients were far from home.

"Didn't want to stay in hospital. Would be better off at home"

"Loneliness is dreadful. Still preferred to come home"

Participants considered discharge planning to be highly individualised. People wanted concrete assurances to ensure that care and equipment would be in place when they arrived home.

"I prefer home care to be in place before you come home - that would be good"

"How do I know that services will be there when I get home?"

Priority 2: Staying in hospital when necessary

As in other studies, patients sometimes felt they were pressurised to leave hospital too soon [23, 24] and wished to stay in hospital for longer until they felt better [25].

"Sometimes you need to be in hospital"

"If you have no visitors at home or family, then it is better to be in hospital for a bit longer"

The issue of assurance and some difficult experiences of the service were expressed, suggesting that the quality of care had caused some anxiety or they felt 'rushed' [23], whilst acknowledging that staff were under pressure to use their bed for another patient [25].

"They were going to discharge me at 10pm, but I refused, and started to bleed later in the night"

"I was drugged and was not clear about what was happening and the Doctor didn't tell me anything and I was discharged although I was feeling pain"

"Once you are ready to go, you want to go because when they are busy you can feel like you are in the way"

Priority 3: Staying independent

Many participants suggested that an important element of further assessment should include functional activity. A majority of participants expressed the desire for future research to prioritise projects investigating ways for the older adults to maintain their independence with daily living at home for as long as possible. Most could see the benefits of being assessed at home and motivated by the idea of personal independence:

"If you don't have home care, assessment at home is a good idea. I got good exercises from them. I liked the idea. They got me back walking"

"Whatever keeps your independence is a good thing. You lose your modesty in hospital."

"Assessment at home was more meaningful than in hospital".

Priority 4: Appropriateness of D2A

There were some specific comments about the discharge process and these ideas were mixed and strongly related to individual social circumstances and personal feelings about what the hospital admission could achieve. There was apparently a shared view that those who were alone would need longer admissions.

"If you live on your own and do not have support you might not like Discharge to Assess"

"Discharge to Assess was good for some but not others"

"Do a home visit to assess the person, and if they can't manage then keep them a bit longer"

Priority 5: Communication processes

All patients expected clear communication between hospital staff and patients/families,. They especially wanted

an opportunity to plan with relatives and carers for returning home.

"Patients and families are not aware of the discharge plan at all"

"Tell the patients and their families about the discharge plan early not just before discharging"

"Sign post who to go to in the community for further advice"

Participants stated that hospital staff sometimes made assumptions about the care that family members provide at home, especially in the BAME community. This could mean that questions related to home care are not asked by discharge planners.

"Nobody asked me if there was help at home"

"Check who is at home and that they are well enough to look after you"

Priority 6: Maintaining research priorities

A more permanent PPIE panel was proposed at the end of interviews/ group sessions; suggesting that this could enable the voice of the older population to shape future research projects. It was thought to be a positive step forward, although none of the participants from this project ultimately became involved when a panel was established.

Discussion

In summary, the frail older patients from across the city presented a range of views on hospital discharge. They assumed that staff would communicate with them in relation to discharge and make sure that social issues would be taken into account, but this was not always the case. The transitions between hospital and home encountered on discharge may represent many challenges but returning 'home' clearly has strong associations with being independent.

Engagement with patients and carers enabled a range of views to be gathered that could inform future research and service development. Results from the D2A project captured important perspectives from frail service users. A key finding was the wish to remain as independent as possible, which aligns to the top research priority identified by people with dementia from the James Lind Alliance [26]. As suggested by previous research findings that explore older adult views [27–29], our findings supported the belief that most people wish to be discharged home as soon as possible (priority 1),

but this was qualified by the desire for the quality of the discharge to be equally as important and tailoring to individual needs is vital [30].

.There were some differences expressed within the participant groups in regard to communication, with members from the BAME community stating that assumptions were made in regard to their home support leading to a potential lack of formal services, something less likely to be stated by non BAME participants. Communication problems in general were expressed by many participants [31] regardless of ethnicity, and reflect similar problems raised by the adoption of D2A within a London Healthcare Trust, although these improved over time [32].

Other findings from the project revealed few differences in the responses of the participants regardless of socio-economic status, ethnicity or gender. The findings confirm some of the concerns raised by patients, including the frail older adults, in other studies. These included the need to take more notice of patient's own perception of when they are ready for discharge, the need for increased formal support on discharge especially for the "older old" and those facing deteriorating life-limiting conditions, and improved partnership working between patients, their family and carers, and health and social care professionals. This applied both pre and post discharge [33–36] and views appear to have changed little over time.

In light of the findings, the challenge for those involved in discharge planning is to take greater account of patient's own concerns, particularly in regard to their perceptions of when they are ready for discharge. The findings have been reported to the clinicians involved in D2A within the Trust enabling positive changes to be made to some of the processes involved with hospital discharge of the frail older people. The D2A participants suggest that future research needs to take account of the effect of earlier discharge on patients to result in a better patient experience [37], the latter being an NHS performance indicator which requires improvement [38]. This is particularly important to ensure the drive for service efficiencies and competing demands for resource allocation within healthcare do not adversely affect patient care and experience [4].

Earlier communication with family and carers may reduce incorrect assumptions made regarding the level of support at home, as well as giving family members reassurance about when their relative is to be discharged and with what level of professional support, allowing them to make necessary preparations themselves. An ongoing challenge is the increased demand for formal support, especially home care, with the increasing proportion of frail older population, at a time of financial restraint in social care budgets [39]. Without an appropriate

increase in support, patients experiencing early discharge are likely to be readmitted [34, 40].

Learning from the PPIE process

Participants for this project mainly attended focus groups despite the literature suggesting most older people prefer to be consulted individually in their own homes [41]. This was considered appropriate where participants lived within the same retirement complex and were therefore able to attend in familiar surroundings without the need to travel. Similarly, members of the PMC and over 50s club were happy to be consulted at their usual venues in a group situation, as they were familiar with their surroundings and each other, and could access them easily. Where people were not able to travel, or preferred not to participate in a group, individual interviews were offered. This was considered essential if a range of older adult voices were to be heard, and was a key consideration when planning future consultations with the participant group.

The PPIE project reinforced the understanding that frail older adults are not a homogenous group and a variety of methods to engage were offered to account for this. The challenge to undertake a sophisticated method of PPIE, requires a senior commitment to funded and co-ordinated activity as a part of research structures within a Trust. This level of commitment to PPIE is not universal. In addition the benefit of having a well-trained and supported PPIE group was evident in the successful recruitment to the D2A project. The advice from the initial PPIE group enabled a wider range of patients with experience of the D2A project to be included, ensuring views that reflected diverse communities. As a direct result of the PPIE project and the D2A public engagement the NHS Trust acknowledged that more permanent public involvement was required; prompting the formation of a sustainable older adult PPIE group, the Community Elders Panel (CEP). Aligned to the guidance provided by INVOLVE (the national advisory group supporting greater public involvement in the NHS) to identify, prioritise, design, conduct and disseminate NHS research [10], the CEP has been developed to support patient engagement and involvement in research, conducted within the Hospital Trust and local region.

The formation of the CEP was enabled by the NHS Trust committing to fund a part time PPIE coordinator. This provided the time and resources to recruit and train potential members and provide ongoing facilitation of the panel. Having succeeded in recruiting a range of participants for the D2A engagement following the advice of the original PPI group, similar methods were followed in recruiting to the CEP. Established community links and networks, as well as clinicians with links to older adult patients in the community were utilised to

ensure a diverse range of panel members were recruited. It was hoped some of those who participated in the D2A engagement would join the CEP but this did not occur. The time frame from the initial project to the start of CEP recruitment was over 3 years which may account for this. Another factor may have been the D2A participants' unfamiliarity with the PPIE coordinator who had not been involved in the original project. However, some of the contacts and venues used to recruit to the D2A project were used to successfully recruit to the panel.

The panel consists of patients and members of the public who are over 75, with some mobility issues and a wide range of health conditions and care needs. The panel is ethnically diverse; both genders are represented as are a wide range of socio-economic backgrounds. Sustainability has been achieved by providing bespoke training packages and consultations taking place in the members own homes, negating the need to travel and ensuring all individuals have the opportunity to be listened to. Visiting volunteer participants individually in their home environment has been shown to be successful in reaching very old and frail people [3].

Impact

Ongoing support from the PPIE coordinator has enabled meaningful engagement with frail older housebound people, ensuring better representation of their views and priorities. Improved design of studies, more realistic outcome measures and research priorities with more relevance to older adults are some of the benefits gained so far from consulting the CEP. One of the research priorities identified by the D2A engagement was investigating methods to enable older adults to remain independent for longer, and the CEP have consulted on several studies involving these themes. The impact on panel members has also been positive; a feeling of contributing again to society, becoming more informed about the work and research being conducted within the Trust and having the opportunity to shape the direction of future healthcare were all benefits expressed by those taking part.

Limitations

The findings from the D2A public engagement may not be generalizable to other areas. Efforts were made to ensure diversity when recruiting to both the D2A engagement and the CEP, however using gatekeepers may have inadvertently excluded some groups. The PPIE and follow-on activity benefitted from an institutional commitment to the formation and maintaince of formal panel activity and co-ordination within a care group. Results of the ongoing panel will be evaluated [42, 43].

Conclusion

Three examples of PPIE with frail older adults have been described, each influencing and inspiring the next. The initial consultation with an existing PPIE group allowed more successful recruitment to the discharge planning (D2A) engagement project, both in terms of numbers and the diversity of those consulted. This engagement had two important outcomes. The first was to deepen the researchers understanding about critical issues for this patient population and the importance of safe and timely discharge from hospital. This is clearly a shared priority for services and for patients and carers. Patients recognised the need to plan their discharge in relation to family, social and community influences and saw the discharge process as an outcome of hospital treatment in relation to their own independence. The second was the perceived importance of learning directly from older people and ensuring their voices are heard. Establishing a sustainable PPIE infrastructure has enabled the views of older adults to be incorporated into the design, delivery and promotion of research projects and service redesign. Careful attention to gaining a range of views, especially from those seldom heard, has the potential to contribute to future success in the growing research programme in elderly medicine both nationally and internationally.

Appendix

Information sheet and topic questions. Patient and Public Involvement Process to investigate the views and experiences of patients and relatives being discharged from hospital.

□We are preparing to undertake some research and would like further information about the patient experience of discharge to inform our research question and design.

□We are aware that the "Discharge to Assess" practice aims to reduce the length of stay in hospital and that frail elderly people and their relatives may have a view about the length of stay and admission to hospital.

□We are also interested in how discharge is experienced by patients and carers of older people and what elements were helpful and what could be improved.

□Please complete the consent documentation saying that you understand that you are free to withdraw at any point and understand that we will be taking notes. Please could you provide feedback on the following;

□What was the experience of staying in hospital: awaiting discharge for some time or leaving hospital quickly?

□What do you think the NHS could learn from your experience?s.

□Do you have a preference in how you would like to be discharged if you were admitted again?

□Do you see longer or shorter hospital admissions as beneficial, and why?. We have a patient panel at the hospital, and would like to know if you would like to continue to help us develop our research questions and comment on proposals.

Abbreviations

BAME: Black, Asian and Minority Ethnic; D2A: Discharge to assess; NHS: National Health Service; PMC: Pakistani Muslim Centre; PPIE: Public Involvement and Engagement; RDS Y&H: Regional Design Service of the NIHR in Yorkshire and Humber; RfPB: Research for Patient Benefit

Acknowledgements

The authors would like to acknowledge Jim Gordon Chair of the Therapeutics and Palliative Care PPI panel for his helpful critique of an early draft of the paper and to all the contributors to the PPIE project.

Funding

£500 of funding was made available retrospectively for the purpose of travel and subsistence for the involvement of older adults by Yorkshire and Humberside NIHR Research and Design service.

Availability of data and materials

Data associated with the project is archived within STH as anonymised records of PPIE activity, under the ownership of DB and may be made available on request.

Authors' contributions

SFD and DB undertook the original PPIE project assisted by IM. AS and LF were responsible for subsequently developing the community elders' panel and all parties contributed drafting, development and final version of the paper. All authors read and approved the final manuscript.

Ethics approval and consent to participate

As a PPIE project, this activity was exempt from ethics approval although was registered with Sheffield Teaching Hospitals Trust for the purpose of governance and reporting.

Consent for publication

All co-authors have reviewed the final draft and have approved submission to this journal.

Competing interests

The authors declare that they have no competing interests.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹Sheffield Hallam University and Sheffield Teaching Hospitals NHS Foundation Trust, Montgomery House, 32 Collegiate Crescent, Sheffield S10 2BR, UK. ²Sheffield Teaching Hospitals NHS Foundation Trust, Sheffield, UK. ³Sheffield Hallam University, Sheffield, UK.

Received: 13 November 2018 Accepted: 12 February 2019 Published online: 21 February 2019

References

- INVOLVE. Briefing notes for researchers: Public involvement in NHS, public health and social care research. http://www.invo.org.uk/wp-content/ uploads/2014/11/9938_INVOLVE_Briefing_Notes_WEB.pdf Accessed 28 Nov 2018
- Faulkner A, Thomas P. User-led research and evidence-based medicine. Br J Psychiatry. 2002;180(1):1–3 https://doi.org/10.1192/bjp.180.1.1.

- Patmore C. Can managers research their own services? An experiment in consulting frail, older community care clients. J Integr Care. 2001;9(5):8–17 https://doi.org/10.1108/14769018200100036.
- Gregory A, Mackintosh S, Kumar S, Grech C. Experiences of health care for older people who need support to live at home: a systematic review of the qualitative literature. Geriatr Nurs. 2017;38(4):315–24 https://doi.org/10.1016/ i.gerinurse.2016.12.001.
- Thilo F, Bilger S, Halfens R, Schols J, Hahn S. Involvement of the end user: exploration of older people's needs and preferences for a wearable fall detection device - a qualitative descriptive study. Patient Prefer Adherence. 2016;11:11–22 https://doi.org/10.2147/PPA.S119177.
- Pratesi A, Sixsmith J, Woolrych R. Genuine partnership and equitable research: working "with" older people for the development of a smart activity monitoring system. Innov J. 2013;18(3):1715–3816 https://www. innovation.cc/scholarly-style/18_3_6_pratesi_partner-equal-research.pdf.
- Hanson E, Magnusson L, Nolan J, Nolan M. Developing a model of participatory research involving researchers, practitioners, older people and their family carers. J Res Nurs. 2006;11(4):325–42 https://doi.org/10.1177/1744987106065829.
- Minkler M. Community-based research partnerships: challenges and opportunities. J Urban Health. 2005;82(2):ii3–12 https://doi.org/10.1093/ jurban/jti034.
- Puts MTE, Toubasi S, Andrew MK, et al. Interventions to prevent or reduce the level of frailty in community-dwelling older adults: a scoping review of the literature and international policies. Age Ageing. 2017;46(3):383–92 https://doi.org/10.1093/ageing/afw247.
- Patmore C, Qureshi H, Nicholas E. Consulting older community care clients about their services: some lessons for researchers and service managers.
 Research policy and planning: the journal of the social services research.
 Group. 2000;18(1):4–11 http://ssrg.org.uk/rpp-18-1-charles-patmore-consulting-older-community-care-clients-about-their-services-some-lessons-for-researchers-and-service-managers/.
- Ismail MM, Gerrish K, Naisby A, Salway S, Chowbey P. Engaging minorities in researching sensitive health topics by using a participatory approach.(report). Nurse Res. 2014;22(2):44 https://doi.org/10.7748/nr.22.2.44.e1268.
- Sixsmith J, Fang ML, Woolrych R, Canham SL, Battersby L, Sixsmith A. Ageing well in the right place: partnership working with older people. Working with Older People. 2017;21(1):40–8 https://doi.org/10.1108/ WWOP-01-2017-0001.
- Cordis. Futurage: A road map for European ageing research. Sheffield: University of Sheffield; 2011. https://cordis.europa.eu/result/rcn/162491_en.html Accessed 16 Nov 2018
- McMurdo MET, Roberts H, Parker S, et al. Improving recruitment of older people to research through good practice. Age Ageing. 2011;40(6):659–65 https://doi.org/10.1093/ageing/afr115.
- Morse A. Discharging older patients from hospital National Audit Office 2016 https://www.nao.org.uk/wp-content/uploads/2015/12/Dischargingolder-patients-from-hospital-Summary.pdf Accessed 16 Nov 2018.
- NICE Transition between inpatient hospital settings and community or care home settings 2016 https://www.nice.org.uk/guidance/qs136/documents/ briefing-paper Accessed 6 Nov 2018.
- Coffey A, McCarthy GM. Older people's perception of their readiness for discharge and postdischarge use of community support and services. Int J of Older People Nurs. 2013;8(2):104–15 https://doi.org/10.1111/j.1748-3743. 2012.00316.x.
- Rojas-García A, Turner S, Pizzo E, Hudson E, Thomas J, Raine R. Impact and experiences of delayed discharge: a mixed-studies systematic review. Health Expect. 2017;21(1):41–56 https://doi.org/10.1111/hex.12619.
- England NH. Five year forward view. 2014. https://www.england.nhs.uk/wp-content/uploads/2014/10/5yfv-web.pdf Accessed 28 Nov 2018.
- Kings Fund. Delayed transfer of care: a quick guide. 2018. https:// www.kingsfund.org.uk/publications/delayed-transfers-care-quick-guide Accessed 28 Nov 2018.
- NIHR. HEE/NIHR clinical doctoral research fellowship. https://www.rds-yh. nihr.ac.uk/public-involvement/rds-yh-publicinvolvement-fund/. Accessed 17 Feb 2019.
- Locock L, Boylan A, Snow R, Staniszewska S. The power of symbolic capital in patient and public involvement in health research. Health Expect. 2017; 20(5):836–44 https://doi.org/10.1111/hex.12519.
- Fuji KT, Abbott AA, Norris JF. Exploring care transitions from patient, caregiver, and health-care provider perspectives. Clin Nurs Res. 2013;22(3):258–74 https://doi.org/10.1177/1054773812465084.

- 24. Sheridan NF, Kenealy TW, Kidd JD, et al. Patients' engagement in primary care: powerlessness and compounding jeopardy. A qualitative study. Health Expect. 2015;18(1):32–43 https://doi.org/10.1111/hex.12006.
- Ekdahl AW, Linderholm M, Hellström I, Andersson L, Friedrichsen M. 'Are decisions about discharge of elderly hospital patients mainly about freeing blocked beds?' a qualitative observational study. BMJ Open. 2012;2(6):e002027 https://doi.org/10.1136/bmjopen-2012-002027.
- James Lind Alliance. Dementia top 10. http://www.jla.nihr.ac.uk/prioritysetting-partnerships/dementia/top-10-priorities/ Accessed 6 Nov 2018.
- NHS England. Monthly delayed transfer of care situation reports definitions and guidance.2015. https://www.england.nhs.uk/statistics/statistical-workareas/delayed-transfers-of-care/. Accessed 17 Feb 19
- Stocker R, Bamford C, Brittain K, et al. Care home services at the vanguard: a
 qualitative study exploring stakeholder views on the development and
 evaluation of novel, integrated approaches to enhancing health care in care
 homes. BMJ Open. 2018;8(3):e017419 https://doi.org/10.1136/bmjopen2017-017419.
- Lévesque L, Ducharme F, Caron C, et al. A partnership approach to service needs assessment with family caregivers of an aging relative living at home: a qualitative analysis of the experiences of caregivers and practitioners. Int J Nurs Stud. 2010;47(7):876–87 https://doi.org/10.1016/j.ijnurstu.2009.12.006.
- Gonçalves-Bradley DC, Lannin NA, Clemson LM, Cameron ID, Shepperd S. Discharge planning from hospital. Cochrane database of systematic reviews 2016, issue 1. Art. No.: CD000313. doi:https://doi.org/10.1002/14651858. CD000313.pub5.
- George S, Atwal A. Discharge planning reduces length of stay and readmission rates for older people admitted with a medical condition. Aust Occup Ther J. 2013;60(5):375–6 https://doi.org/10.1111/1440-1630.12087.
- Meehan L, Banarsee R, Dunn-Toroosian V, Tejani S, Yazdi A. Improving outcomes for patients discharged early using a home assessment scheme. London J Prim Care. 2018;10(3):62–7 https://doi.org/10.1080/17571472.2018. 1489467.
- Reay PA, Horner B, Duggan R. The patient's experience of early discharge following total hip replacement. Int J Orthop Trauma Nurs. 2015;19(3):131–9 https://doi.org/10.1016/j.iiotn.2015.02.003.
- Allen J, Hutchinson AM, Brown R, Livingston PM. User experience and care for older people transitioning from hospital to home: Patients' and carers' perspectives. Health Expectations. 2018;21(2):518–27. doi: https://doi.org/10. 1111/hex.12646.
- Clarke A, Sohanpal R, Wilson G, Taylor S. Patients' perceptions of early supported discharge for chronic obstructive pulmonary disease: a qualitative study. BMJ Qual Saf. 2010;19(2):95–8 https://doi.org/10.1136/ qshc.2007.025668.
- Mamon J, Steinwachs DM, Fahey M, Bone LR, Oktay J, Klein L. Impact of hospital discharge planning on meeting patient needs after returning home. Health Serv Res. 1992;27(2):155 https://www.health.org.uk/publications/ measuring-patient-experience.
- The Health Foundation. Measuring patient experience.; evidence scan. The Health Foundation. 2013. https://www.health.org.uk/publications/measuringpatient-experience. Accessed 17 Feb 19
- Department of Health. NHS outcomes framework: at a glance. https://assets. publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/513157/NHSOF_at_a_glance.pdf. Accessed 17 Feb 19
- Shankar A, McMunn A, Demakakos P, Hamer M, Steptoe A. Social isolation and loneliness: prospective associations with functional status in older adults. Health Psychol. 2017;36(2):179 https://doi.org/10.1037/hea0000437.
- Slatyer S, Toye C, Popescu A, Young J, Matthews A, Hill A, Williamson DJ. Early re-presentation to hospital after discharge from an acute medical unit: perspectives of older patients, their family caregivers and health professionals. J Clin Nurs. 2013;22(3–4):445–55 https://doi.org/10.1111/jocn.12029.
- Kitzinger J. The methodology of focus groups: the importance of interaction between research participants. Sociol Health Illn. 1994;16(1):103–21 https:// doi.org/10.1111/1467-9566.ep11347023.
- Gordon J, Franklin S, Eltringham SA. Service user reflections on the impact of involvement in research. Res Involv Engagem. 2018;4(1):11 https://doi. org/10.1186/s40900-018-0095-1.
- Mockford C, Staniszewska S, Griffiths F, Herron-Marx S. A systematic review of the impact of patient and public involvement on NHS healthcare services. http://oro.open.ac.uk/38582/. Accessed 17 Feb 19

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

